Author’s response to reviews

Title: 1-1-8 one-step sevoflurane wash-in scheme for low-flow anesthesia: simple, rapid, and predictable induction

Authors:
Sririrat Tribuddharat (sirirat.tribuddharat@gmail.com)
Thepakorn Sathitkarnmanee (thepakorns@gmail.com)
Naruemon Vattanasiriporn (anny_nana@hotmail.com)
Maneerat Thananun (t.manee03@gmail.com)
Duangthida Nonlhaopol (danon401@gmail.com)
Wilawan Somdee (worapon37@gmail.com)

Version: 2 Date: 12 Jan 2020

Author’s response to reviews:

Response to Editor’s comments

Editor Comments:
1. Trial registration

-- As your study was registered after enrollment of the first participant, please include the words 'retrospectively registered' in your trial registration statement at the end of the Abstract.
Response: We included ‘retrospectively registered' at the end of the Abstract (Page 3; Line5)

2. Availability of data and materials

-- The Availability of data and materials section refers to the raw data used in your study and presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files. Please only use this statement if you have indeed provided all raw data on which your study is based.
Response: We make change to “The data used to support the findings of this study are available from the corresponding author upon request.” (Page 12; Line 11-13)

3. CONSORT

-- CONSORT refers to recommendations for reporting randomised trials. As this is not applicable to your current study, as it is not randomised, please remove the statement regarding CONSORT in your Methods section.
Response: We removed this statement in Methods section.
4. Additional files

-- Please remove the response to the reviewers’ comments, cover letter, and CONSORT files from the file inventory as these are no longer needed at this stage of the editorial process.
Response: We followed your suggestion.

5. Duplicate tables

-- Please remove the duplicated tables in your manuscript.
Response: We removed the duplicated Tables in our manuscript.

6. Clean copy

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
Response: We followed your suggestion.